Author's response to reviews

Title: Prevalence of distressing symptoms in hospitalised patients on medical wards. A cross-sectional study.

Authors:

Katrin R Sigurdardottir (katrin.sigurdardottir@haraldsplass.no)
Dagny F Haugen (dagny.haugen@ntnu.no)

Version: 4 Date: 11 September 2008

Author's response to reviews: see over
Author’s response to reviews

Title:
Prevalence of distressing symptoms in hospitalised patients on medical wards. A cross-sectional study.

Authors:
Katrin Ruth Sigurdardottir (katrin.sigurdardottir@haraldsplass.no)
Dagny Faksvåg Haugen (dagny.haugen@ntnu.no)

Version: 3 Date: 11 September 2008

Author’s response to reviews:

Dear Sir/Madam

I am submitting a revised version of the manuscript which incorporates the formatting changes as requested.

Kind regards

Dr Katrin Sigurdardottir (11 September 2008)